Author’s response to reviews

Title: Botulinum Toxin Injection versus Lateral Internal Sphincterotomy in the treatment of Chronic Anal Fissure: a non-randomized controlled trial.

Authors:

Dr Adnan Giral (adnangiral@yahoo.com)
Kemal Memisoglu (kmemisoglu@hotmail.com)
Yucel Gultekin (kkmemisoglu@hotmail.com)
Nese Imeryuz (nimeryuz@yahoo.com)
Cem Kalayci (ckalayci@superonline.com)
Prof Nefise B Ulusoy (nefiseulusoy@hotmail.com)
Nurdan Tozun (nurdantozun@marmara.edu.tr)

Version: 2 Date: 29 Feb 2004

Dear Editor,

We made discussion and reference updating according to Mr. Jost and Mr. Brisinda's advices. Language corrections are also made. Intergroup comparisons were made but this is come to clear on the table 2 in this revision.

Best regards,

Adnan Giral M.D.

Marmara University
School of Medicine
Department of Gastroenterology
Istanbul Turkey